Responses

Title: Risk factors for Contracting Watery Diarrhoea in Kadoma City, Zimbabwe, 2011: A Case Control Study

1. page 3. BACKGROUND

Please insert ‘Figure 1’ as shown:

In 2011, Kadoma City experienced an upsurge of watery diarrhoeal cases during week 36, in 2011 (Figure 1). The total cases.

We have inserted ‘Figure 1’ as advised.

2. page 8: Descriptive Epidemiology Replace semicolon by comma

People, whilst for those above 5 years old; the attack rate.

We have replaced semicolon by comma.

3. pg10. Knowledge of Caregivers. The knowledge on prevention and home treatment of diarrhoea was fair, and did not differ significantly between amongst the cases and controls as shown in Table 2. Please, delete amongst.

We have deleted amongst as advised.

4. Please correct the bacteria name to E. coli or Escherichia coli, since you have incorrectly used capital letter for the species name (i.e., E. Coli).

We have corrected the bacteria name and now reads Escherichia coli throughout the document.

Competing Interests:
Please be advised that manuscripts must include a *Competing interests?* section. This should be placed after the Conclusions/Abbreviations. If there are none to declare, please include the statement *The authors declare that they have no competing interests.* Please consider the following questions and include an appropriate declaration of competing interests in your manuscript:

We have included a Competing interests section and since there are none to declare we have included the statement ‘The authors declare that they have no competing interests’